Author’s response to reviews

Title: A model to predict unstable carotid plaques in population with high risk of stroke

Authors:

Junxiong Yin (xiong201@163.com)
Chuanyong Yu (442855931@qq.com)
Hongxing Liu (maclhx@yahoo.com)
Mingyang Du (dmyang0128@126.com)
Feng Sun (15506264@qq.com)
Cheng Yu (289032407@qq.com)
Lixia Wei (434598706@qq.com)
Chongjun Wang (1391396536@139.com)
Xiaoshan Wang (professor_wxs@163.com)

Version: 4 Date: 18 Mar 2020

Author’s response to reviews:

Dear Editor,

We thank you for the comments, which helped to improve our study. Based on these comments and suggestions, we have made careful modifications on the original manuscript. We made the revisions as below:

1. We removed the funding information from the Acknowledgements section to the Funding section.

2. We statement the role of the funding body. They had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. We added a section "Additional files", listed the information for each additional files. We cited all figures, tables and additional files within the text.

4. We made the manuscript be a clean version, which does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files also are clean versions.

5. We have modified the format according to the latest requirements.
We hope that these revisions are satisfactory and that the revision will be acceptable for publication in “BMC Cardiovascular Disorders”.

Thank you very much for your work concerning our paper.

Wish you all the best!

Sincerely yours,
Xiaoshan Wang